Author's response to reviews

Title: The effects of the Two-Week Rule on NHS colorectal cancer diagnostic services: A systematic literature review

Authors: Kymberley Thorne (k.thorne@swansea.ac.uk)  
Hayley A Hutchings (h.a.hutchings@swansea.ac.uk)  
Glyn Elwyn (elwyn@cardiff.ac.uk)

Version: 4 Date: 27 March 2006

Author's response to reviews:

An acknowledgement section has been added at the end of the paper